\documentclass[12pt]{article}
\usepackage{amsmath}
\setlength\parindent{0pt}

%PAGE LAYOUT
\pagestyle{empty}
\setlength{\topmargin}{-1in}
\setlength{\headheight}{1.5cm}
\setlength{\headsep}{0.3cm}
\setlength{\textheight}{9in}
\setlength{\textwidth}{6.5in}
\setlength{\oddsidemargin}{0cm}
\setlength{\evensidemargin}{0cm}

%\usepackage{graphicx}

\begin{document}

\begin{tabular}{@{}l p{4.5cm} r@{}}
	Name: \hrulefill & & Math 1441 Quiz 10 \\[0.2cm]
	\hspace{8cm} & & 
\end{tabular}

{\bf 1.} (6 points) Find the area of the region enclosed by the parabola $y=x^2$ and the line $y=4$. \\[5cm]

{\bf 2.} (6 points) Evaluate the indefinite integral by using the given substitution.
$$\int 2x \cos(x^2) \, dx,\quad u=x^2. $$\\[5cm]

{\bf 3.} (8 points) Evaluate the indefinite integral.
$$\int \frac{x}{(1+x^2)^2}\, dx$$\\[5cm]

\end{document}